

### CRISPR/Cas9 -mediated gene knockout of Anopheles gambiae FREP1 suppresses malaria parasite infection

Yuemei Dong, Maria L Simões, Eric Marois, George Dimopoulos

#### ▶ To cite this version:

Yuemei Dong, Maria L Simões, Eric Marois, George Dimopoulos. CRISPR/Cas9 -mediated gene knockout of Anopheles gambiae FREP1 suppresses malaria parasite infection. PLoS Pathogens, 2018, 14 (3), pp.e1006898. 10.1371/journal.ppat.1006898 . inserm-02171043

### HAL Id: inserm-02171043 https://inserm.hal.science/inserm-02171043

Submitted on 2 Jul 2019

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



### 

**Citation:** Dong Y, Simões ML, Marois E, Dimopoulos G (2018) CRISPR/Cas9 -mediated gene knockout of *Anopheles gambiae FREP1* suppresses malaria parasite infection. PLoS Pathog 14(3): e1006898. https://doi.org/10.1371/ journal.ppat.1006898

Editor: Kenneth D Vernick, Institut Pasteur, FRANCE

Received: November 14, 2017

Accepted: January 23, 2018

Published: March 8, 2018

**Copyright:** © 2018 Dong et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the paper and its Supporting Information files.

**Funding:** This work was supported by the National Institutes of Health / National Institute of Allergy and Infectious Disease grants R21Al131574 (to GD) and R01Al122743 (to GD), and a Johns Hopkins Malaria Research Institute Postdoctoral Fellowship (to MLS), the Bloomberg Philanthropies (to GD). CNRS, Inserm, the University of Strasbourg and Agence Nationale de la Recherche RESEARCH ARTICLE

### CRISPR/Cas9 -mediated gene knockout of *Anopheles gambiae FREP1* suppresses malaria parasite infection

#### Yuemei Dong<sup>1</sup>, Maria L. Simões<sup>1</sup>, Eric Marois<sup>2</sup>, George Dimopoulos<sup>1</sup>\*

1 W. Harry Feinstone Department of Molecular Microbiology and Immunology, Bloomberg School of Public Health, Malaria Research Institute, Johns Hopkins University, Baltimore, Maryland, United States of America, 2 Inserm, CNRS, Université de Strasbourg, Strasbourg, France

\* gdimopo1@jhu.edu

### Abstract

*Plasmodium* relies on numerous agonists during its journey through the mosquito vector, and these agonists represent potent targets for transmission-blocking by either inhibiting or interfering with them pre- or post-transcriptionally. The recently developed CRISPR/Cas9based genome editing tools for *Anopheles* mosquitoes provide new and promising opportunities for the study of agonist function and for developing malaria control strategies through gene deletion to achieve complete agonist inactivation. Here we have established a modified CRISPR/Cas9 gene editing procedure for the malaria vector *Anopheles gambiae*, and studied the effect of inactivating the fibrinogen-related protein 1 (*FREP1*) gene on the mosquito's susceptibility to *Plasmodium* and on mosquito fitness. *FREP1* knockout mutants developed into adult mosquitoes that showed profound suppression of infection with both human and rodent malaria parasites at the oocyst and sporozoite stages. *FREP1* inactivation, however, resulted in fitness costs including a significantly lower blood-feeding propensity, fecundity and egg hatching rate, a retarded pupation time, and reduced longevity after a blood meal.

#### Author summary

The causative agent of malaria, *Plasmodium*, has to complete a complex infection cycle in the *Anopheles gambiae* mosquito vector in order to reach the salivary gland from where it can be transmitted to a human host. The parasite's development in the mosquito relies on numerous host factors (agonists), and their inhibition or inactivation can thereby result in suppression of infection and consequently malaria transmission. The recently developed CRISPR/Cas9-based genome editing tools for *Anopheles* mosquitoes provide new and promising opportunities to delete (inactivate) *Plasmodium* agonists to better understand their function and for blocking malaria transmission. Here we have established a modified CRISPR/Cas9 genome editing technique for malaria vector *A. gambiae* mosquitoes. Through this approach we have inactivated the fibrinogen-related protein 1 (FREP1) gene, via CRISPR/Cas9 genome editing, and the impact of this manipulation on the

(grants JCJC "GEMM" and #ANR-11-EQPX-0022) to EM. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

mosquito's susceptibility to *Plasmodium* and on mosquito fitness. *FREP*1 knockout mutants showed a profound suppression of infection with both human and rodent malaria parasites, while it also resulted in fitness costs: a significantly lower blood-feeding propensity, fecundity and egg hatching rate, and a retarded larval development and pupation time, and reduced longevity after a blood meal.

#### Introduction

Malaria remains one of the most devastating infectious diseases, killing nearly half a million people annually [1]. For successful transmission, the malaria parasite has to complete a complex journey in the mosquito vector that involves intimate molecular interactions with the vector's midgut, hemolymph, and salivary glands [2]. These crucial *Anopheles-Plasmodium* interactions represent targets for transmission blocking by inhibiting parasite agonists (or host factors) that are required for infection [3–6]. An obvious advantage of studying *Plasmodium* agonists for malaria control is that they can be targeted through multiple means, such as agonist gene deletion, RNAi-mediated gene silencing, or interference through antibodies or small molecules [5, 7–8]. However, the exploration of mosquito *Plasmodium* agonists for the development of malaria control strategies based on parasite suppression has lagged behind other approaches because of the lack of effective gene editing tools for *Anopheles* mosquitoes. The recently developed CRISPR/Cas9-based genome editing methodology and gene-drive systems for *Anopheles* mosquitoes provide new and promising tools for the study of *Plasmodium* host factors, with regard to their biology and potential for the development of transmission-block-ing strategies.

Type II clustered regularly interspaced short palindromic repeats (CRISPR) bacterial immune systems, together with CRISPR-associated protein 9 (Cas9), have been developed into a powerful tool for genome engineering, and CRISPR/Cas9-based gene editing has been widely explored for biomedical research and therapeutics. This technology has recently also been applied to genome editing in *Aedes aegypti, Anopheles stephensi*, and *Culex* mosquitoes [9–14]. Furthermore, Kistler and colleagues [11] have established a comprehensive protocol for CRISPR/Cas9 gene editing in *Aedes* mosquitoes through embryonic delivery of *in vitro*-synthesized guide RNA (sgRNA) and recombinant Cas9 protein. CRISPR/Cas9 mediated somatic disruption of a male-determining gene in *Aedes* mosquitoes has produced males with feminized genitalia [12]. Cas9-mediated gene drive technology has proven promising for population modification of both Asian and African malaria vector mosquitoes, *A. stephensi* and *A. gambiae*, respectively [15–17].

Here we investigated the effect of fibrinogen-related protein 1 (FREP1) gene inactivation, obtained by CRISPR/Cas9 genome editing, on both susceptibility to *Plasmodium* and mosquito fitness. FREP1 belongs to the fibrinogen-related protein family (FREP, also known as fibrinogen domain immunolectin [FBN]), representing the largest immune- and pattern recognition gene family in *A. gambiae*, with 59 putative members. We have previously shown that several members of the FREP/FBN family are infection-responsive upon challenge with bacteria or *Plasmodium*, and FBN8 (FREP57), FBN9 (FREP13), and FBN39 (FREP40) are involved in anti-*Plasmodium* defense [18–19]. Transgenic overexpression of *FBN9* (*FREP13*) in *A. gambiae* mosquitoes results in an elevated resistance to infections with bacteria and the rodent malaria parasite *P. berghei* [20].

Whereas several FREP members are *Plasmodium* antagonists, FREP1 functions as broadspectrum *Plasmodium* agonist (host factor) [21–23]. FREP1 plays a potent role in the *Plasmodium* ookinete's invasion of the mosquito midgut epithelium [21]. RNAi-based studies have shown that gene silencing of *FREP*1 results in a suppression of *P. falciparum* development in the midgut tissue [21]. Polyclonal anti-FREP1 antibodies can block infection of *Anopheles* with *P. falciparum*, *P. vivax*, and *P. berghei*, suggesting that FREP1 might be a promising transmission-blocking vaccine (TBV) target [23]. While such a TBV has already been developed, here we have taken a different interference approach, using *FREP1* gene disruption to investigate its biological role as a host factor and its potential as a transmission-blocking target for the development of a novel malaria control strategy.

To generate CRISPR/Cas9-based *FREP1* knockout *A. gambiae* mosquitoes, we established a modified CRISPR/Cas9 gene editing method based on crossing a guide RNA (gRNA)-expressing transgenic line with a transgenic line expressing Cas9 under the control of the germlinespecific *Vasa2* promoter [24]. Mosquitoes carrying the desired knockout mutation were isolated in the progeny. The resulting *FREP1* knockout mutant mosquitoes showed a significant decrease in their permissiveness to *P. falciparum* and *P. berghei* infection. However, the mutants also exhibited a significant fitness cost in terms of blood-feeding propensity, longevity after blood meal, fecundity, and egg hatching rates, but no impact on the adult mosquito size or on the life-span of mosquitoes maintained on a 10% sugar solution.

#### Results

#### Generation and characterization of FREP1 knockout mutants

In our preliminary attempts to generate CRISPR/Cas9 knockout null mutants we tried the method established for *Aedes* and *Anopheles stephensi* mosquitoes [11, 15] by co-injecting synthesized gRNAs (sgRNAs) and Cas9 protein into the embryos of *A. gambiae* G3 mosquitoes. The mosquito embryo microinjection was carried out according to established protocols [25–28]. In total, approximately 10% of the eggs hatched from 1000 micro-injected eggs, and the emerged  $G_0$  adults were allowed to outcross with wild type mosquitoes to produce  $G_1$  eggs followed by screening for germline transmission of stable mutations, but not a single mutant was identified in the survivors' progeny.

We then proceeded with the development of a modified method to generate CRISPR/ Cas9-induced knockout mutant in A. gambiae mosquitoes through combination and modification of established approaches [16, 28]. We first created a transgenic A. gambiae line expressing three gRNAs targeting the FREP1 gene (S1 Fig). All three FREP1-targeting gRNAs were driven by the A. gambiae U6 snRNA polymerase III promoter (AGAP013557). The specific sequences of individual gRNAs were synthesized as short DNA linkers (S1 Table) and separately cloned into plasmid modules carrying the U6 promoter sequence followed by a gRNAexpression template, and assembled by a Golden Gate cloning reaction into the pDSAT transgenesis vector [28] containing the 3xP3-TFP reporter (Figs 1A and S1). Upon microinjection into embryos of the A. gambiae X1 docking line [28] and backcrossing of injected survivors to the parental X1 line, transgenic progeny showing stable blue fluorescence in the larvae or adult eyes were isolated for further studies (Fig 1B and 1C). After amplifying the FREP1gRNA-expressing (FREP1-gRNA) transgenic population, a subpopulation enriched with homozygous transgenic mosquitoes was crossed with the Vasa-Cas9 strain [15-16] to generate germ-line gene-knockout mutants. Males from this cross carrying both transgenes were selected by their fluorescent markers (Fig 1D and 1E), and outcrossed to wild type (wt) virgin females (S1 Fig). Only the virgin female progeny from this cross (being potential heterozygous mutants) was crossed with wt males. These females were blood-fed and isolated in individual containers to lay eggs. The mutation region from the mother was assessed by PCR with flanking primers (S1 Table) and the mutation/deletion was confirmed by PCR first (Fig 1F)



**Fig 1. Generation of CRISPR/Cas9-mediated** *FREP1* gene knockout mutants. (A) Schematic representation of the pDSAT-FREP1-gRNA<sub>3</sub> (with blue fluorescence eye reporter gene) transformation plasmid used for the germline transformation of the *A. gambiae* X1 docking line. Three gRNA target DNA sequences of FREP1 were first independently cloned into gRNA expression template vectors and assembled into a pDSAT vector via the Golden Gate cloning technique using *BsaI* sites. (**B**, **C**) Generation of *FREP1-gRNA*-expressing (FREP1-gRNA) transgenic line. Fluorescent images of larvae of FREP1-gRNA (blue) transgenic line, along with the wt control (X1, non-fluorescent) strain (**B**). Fluorescent images of adult transgenic FREP1-gRNA mosquito and wt control (**C**). (**D**, **E**) FREP1-gRNA virgin females were crossed with *Vasa-Cas9* (Cas9, YFP, yellow/green fluorescence) males to generate the transheterozygotes (FREP1/Cas9) for producing *FREP1* gene knockouts. The transheterozygous progeny has both blue and green fluorescence in larval (**D**) and adult eyes (**E**), whereas progeny that did not inherit FREP1-gRNA has only the green fluorescent eye marker. (**F**) PCR validation of *FREP1* gene disruption (sequences available in SI <u>S1 File</u>). 478 bp PCR products from X1, Cas9, FREP1-gRNA, and FREP1/Cas9 transheterozygotes were sequenced (upper panel). PCR showed that the heterozygous *FREP1*-mutants have both wild type band (478 bp) and the lower band (323 bp) with a 155 bp gene deletion (FREP1-KO'). Homozygous deletion mutant (FREP1-KO) showed only one lower band (SI <u>S1 File</u>).

followed by sequencing (SI <u>S1 File</u>). In the progeny from these founder females, PCR on leg DNA was used to screen individual mutant mosquitoes, and all the mutant mosquitoes with the same heterozygous mutation genotype as revealed by PCR and sequencing were pooled. To enrich for homozygous mutants, the whole process was repeated in the next generation to generate homozygous deletion mutants (Fig 1F). The reading frames of *FREP1 gene* were shifted and no FREP1 protein was produced in *FREP1*-KOs (SI S1 File).

# Impact of *FREP1* knockout on susceptibility to the human malaria parasite *P. falciparum*

To assess the effect of the *FREP1* gene knockout on *P. falciparum* infection, we fed the pool of homozygous *FREP1* mutants (to avoid any bias directed to certain single progenies), along with the X1, *Vasa-Cas9* (Cas9), and FREP1-gRNA strains as controls, on a NF54 *P. falciparum* gametocyte culture. Only fully engorged mosquitoes were included in the assay to ensure that the resulting differences in infection were derived from the *FREP1* gene knockout and not from potential changes in feeding behavior (discussed below). *FREP1* knockout mutants (FREP1-KOs) displayed significantly lower permissiveness to pre-oocyst stage *P. falciparum* when fed on blood with a high gametocytemia (0.1%), which resulted in an unnaturally high infection intensity (median of 91 oocysts per mosquito) and prevalence of 98.1% in wild-type (wt) mosquitoes (Fig 2A, S2 Table). The prevalence of infected mosquitoes (those with at least one oocyst on the midgut) decreased by 22.3% (chi-squared test, p<0.0001, S2 Table), and the median infection intensity was reduced by 81.3% when



**Fig 2. Suppression of** *P. falciparum* in *FREP1* gene knockout mutants. (A, B) *P. falciparum* (NF54) oocyst infection intensities of *FREP1* knockout mutants at 8 days post-infection (dpi) when fed on blood with a high (0.1%) (A) or low (0.01%) (B) gametocytemia. *P. falciparum* oocyst infection intensity at 8 dpi in the wt (X1), *Vasa-Cas9* (Cas9), and FREP1-gRNA lines, and *FREP1* knockout mutants (FREP1-KOs). The wt (X1), Cas9, and FREP1-gRNA lines served as controls. (C) The prevalence of the oocyst infection was significantly lower in the *FREP1* knockout mosquitoes at low infection level. At low infection level, *P. falciparum* sporzoite loads in the salivary glands (SG) at 14 dpi (D) and the prevalence of the sporzoites (E) was significantly lower in the *FREP1* knockout mosquitoes. Assays were performed with at least three biological replicates, except for (A) (two replicates), and equal numbers of parasites from different replicates were pooled for the dot-plots. Each dot represents the number of parasites in an individual midgut or salivary gland, and the horizontal lines (red) indicate the median values. Mann-Whitney and Kruskal-Wallis (KW) tests were used to calculate *p*-values and determine the significance of parasite numbers. A chi-squared test was used to compare infection prevalence values. Detailed statistical analysis is presented in S2 Table.

compared to wt X1 control mosquitoes (Kruskal-Wallis test, p<0.0001, Fig 2A, S2 Table). When the mosquito cohorts were fed on a 10-fold lower gametocytemia (0.01%), to mimic a natural infection system, the *FREP1* mutants displayed strong parasite suppression, with a median oocyst count of 0, as compared to the control mosquitoes that had a median oocyst count of 2.0 (X1 and Cas9 strains) and 2.5 (FREP1-gRNA strain). The *FREP1* knockout mutants (FREP1-KOs) displayed an infection prevalence of 41.3%, in contrast to 82.9% for the control X1 mosquitoes (Fig 2B and 2C and S2 Table). At low infection level, *FREP1* gene knockout also resulted in a profound decrease of the sporozoite-stage parasites in the salivary gland (Fig 2D), with a median sporozoite count of 0. The sporozoite infection prevalence showed a 1.9-fold decrease in the *FREP1* knockout mosquitoes (82.1% in the control mosquitoes vs 42.9% in the FREP1-KOs) (Fig 2E, S2 Table).



**Fig 3. Suppression of** *P. berghei* **in CRISPR/Cas9-mediated** *FREP1* **gene knockout mutants. (A, B)** *P. berghei* (wt, ANKA 2.34) oocyst infection intensities of FREP1 knockout mutants (FREP1-KOs) at 13 days post-infection when fed on *P. berghei*-infected mice at high (**A**) or low (**B**) infection level. wt (X1), Cas9, and FREP1-gRNA lines served as controls. (**C**) At low infection level the prevalence of the oocyst infection was significantly lower in the *FREP1* knockout mosquitoes. At least three biological replicates were included in each assay, and equal numbers of mosquitoes from the different replicates were pooled for the dot-plot analysis. Each dot represents the number of parasites in an individual mosquito, and the horizontal lines (red) indicate the medians. *p*-values were calculated using Mann-Whitney and Kruskal-Wallis (KW) tests. A chi-squared test was used to compare infection prevalence values. Detailed statistical analysis is presented in S3 Table.

## Impact of *FREP*1 knockout on susceptibility to the rodent malaria parasite *P. berghei*

To assess the permissiveness of *FREP1* knockout mosquitoes for the rodent malaria parasite *P. berghei*, we fed the knockout mosquitoes, along with the three control group mosquitoes (X1, Cas9, FREP1-gRNA), on *P. berghei* (wt, ANKA 2.34)-infected Swiss Webster mice. Only fully engorged mosquitoes were included in the assay to ensure that the differences observed in infection were related to the *FREP1* gene knockout and not to any change in feeding behavior (discussed below). Gene knockout of *FREP1* resulted in a significantly lower permissiveness to *P. berghei* oocyst infection: a 79.5% and 100% reduction in median infection intensity at high and low infection levels, respectively, when compared to the X1 control mosquitoes (Kruskal-Wallis test, p<0.0001, Fig 3A and 3B, S3 Table). At the high Infection level, the prevalence was 68.8% in the *FREP1* knockouts, vs. 97.5% for the control mosquitoes, while the infection prevalence showed a 2.1-fold decrease from 81.3% in the X1 control mosquitoes to 38.9% in the *FREP1* mutants at the low infection level (chi-squared test, p<0.0001, Fig 3C, S3 Table).

#### Fitness impact of FREP1 gene knockout

To determine whether *FREP1* knockout influences the blood-feeding ability of mosquitoes, we first exposed the *FREP1* knockout mutants and the control (X1, Cas9, FREP1-gRNA) strains to blood-containing membrane feeders, or anesthetized mice, for at least 45 min. All the mosquitoes had been starved of sugar for 3~5 h prior to the assays. *FREP1* knockout mosquitoes showed a 2.6-fold lower feeding propensity when compared to the controls; only 28% of the *FREP1* knockout mosquitoes fed on the human blood-containing membrane feeders, in contrast to 73% of the wt (X1) control mosquitoes (Fig 4A). Similarly, *FREP1* knockout mosquitoes showed a 1.3-fold lower feeding propensity when exposed to mice, with 68% of the *FREP1* knockout mosquitoes feeding on the mice, as compared to 90%~95% of the X1 control mosquitoes (Fig 4B).



**Fig 4. Fitness effects of** *FREP1* **gene knockout.** (**A**, **B**) Percentage of mosquitoes fed on human blood through membrane (**A**) or naïve mice (**B**). A chi-squared test was used to compare blood feeding prevalence values. (**C**) *FREP1* gene knockout (KO) larva developed much more slowly than did control larvae. (**D**) The pupation time of the *FREP1* knockout mutants (*FREP1*-KOs) was significantly slower than that of the wt (X1), *Vasa-Cas9* (Cas9), or FREP1-gRNA control lines. The pupation time lagged behind the control time by a median of 2 days. (**E**) The wing lengths of the *FREP1* mutant females or males did not differ from those of the control mosquitoes. (**F**, **G**) Life spans of the *FREP1* mutants maintained on 10% sucrose solution (**F**) or after one mouse blood meal (**G**). The life spans of the mutants were significantly shorter than those of the controls when the mosquitoes were fed on the naïve mice. The pooled values from three replicates are shown, with standard error bars. Survival rates were analyzed by Kaplan-Meier survival analysis. (**H**) Numbers of eggs laid by female *FREP1* heterozygous mutants (FREP1-KO') or homozygous mutants (FREP1-KOs) were significantly lower than those of the control transgenic lines. Each dot represents the eggs laid by an individual female after a single blood meal on mice. The median values (black horizontal bars) are shown. The *p*-values were calculated with a Mann-Whitney test. (**I**) Hatch rates indicate the average percentage of eggs giving rise to 1<sup>st</sup> and 2<sup>nd</sup> instar larvae, as determined by two replicates from two consecutive generations. Mean values for hatch rates and standard errors (SE) of replicates are indicated. \*\*\*\*\*, p<0.0001.

Next we compared the larval development and pupation time of the *FREP1* knockout mutants to that of the control strain mosquitoes and observed a retarded larval development of the *FREP1* knockout mutants. When 90% of the control mosquito larvae had reached the L4 stage, all of the knockout mutants were still at L2 stage (Fig 4C). The average pupation time of the knockout mutants was 2.0 days longer than that of the control strains (Kruskal-Wallis test, p<0.0001) (Fig 4D). However, the adult *FREP1* knockout mosquitoes that emerged showed no difference in their body size from those of the control strains, as measured by the wing lengths in both female and male mosquitoes (Fig 4E). The life span of the adult *FREP1* knockout mosquitoes, when mosquitoes were maintained on a 10% sugar solution in the absence of a blood meal (Fig 4F). Exposure of mosquitoes to a single naïve blood meal resulted in an increased longevity of the control mosquitoes but not of the *FREP1* knockout mosquitoes (Kaplan-Meier survival analysis, p<0.0001, Fig 4G).

Fecundity, as a measure of egg-laying capacity, was significantly lower for the heterozygous and homozygous *FREP1* knockout mosquitoes than for the controls (Mann-Whitney test, p<0.0001, Fig 4H). The egg hatch rate was also significantly lower for the heterozygous and homozygous *FREP1* mutants (chi-squared test, p<0.0001, Fig 4I).

#### Discussion

Here we have, for the first time, created an A. gambiae Plasmodium host factor gene knockout mosquito using CRISPR/Cas9 genome editing. We targeted the mosquito-encoded Plasmodium host factor FREP1, which plays an essential role during the parasite's midgut infection stage. FREP1 had previously been investigated for its role as a *Plasmodium* host factor by using RNAi-mediated gene silencing methodology and FREP1-inhibiting polyclonal antibodies, and these studies showed that it represents a host factor for P. falciparum, P. vivax, and P. berghei. However, FREP1 silencing resulted in an average 50% and 11% decrease in P. falciparum infection intensity and prevalence, respectively, at a high infection level (with median oocyst counts of 50 and 20 in two replicates) [21]. The complete inactivation of the FREP1 gene using gene editing showed a much stronger suppression of Plasmodium, most likely because RNAi-based gene silencing results in only partial protein depletion. The refractoriness level of FREP1 knockout mosquitoes to P. falciparum was similar to that achieved when the anti-Plasmodium IMD pathway transcription factor *Rel2* was transgenically over-expressed after a blood meal, or when the insulin pathway was transgenically activated [25, 29]. Parasite suppression at the sporozoite stage in the FREP1 knockout mosquitoes suggests that an epidemiologically significant impact would be achieved if these mosquitoes would replace wt mosquitoes and were able to persist in a malaria-endemic region [30-32].

Significant advances in the development of A. gambiae gene drives has been achieved, aimed at the development of mosquito population replacement methods in which a mosquito mutation, or transgene, causing resistance to the parasite could be propelled and spread into a population of susceptible mosquitoes. However, depending on the efficiency of a gene-drive mechanism, a strong fitness impact of the mutations and transgenes could make this approach unfeasible for spreading transgenic malaria-resistance into a mosquito population [33]. Mosquito host factors for *Plasmodium* often play essential biological roles, and their complete inactivation at all developmental stages and in all tissues would therefore be expected to adversely affect fitness. Consistent with this expectation, the FREP1 knockout mosquitoes showed a diminished feeding propensity, a slower pre-adult development, and decreased fecundity; furthermore, a blood meal failed to improve their longevity. All these fitness effects may be related to FREP1's role in blood feeding and digestion as a putative peritrophic matrix factor [21] since its inactivation could result in decreased nutritional uptake or in impaired midgut protection against stress. The peritrophic matrix is a key regulator of mosquito gut homeostasis [34] and nutritionally deprived female mosquitoes may lay fewer eggs with decreased hatching rate and prolonged larval developing time. Several fibrinogen (FBG) domain-containing proteins have been shown to play roles in processes taking place in the blood. For example, the vertebrate fibrinogen (FBG) proteins, ficolins, are involved in blood clotting [35–36]. The function of FREPs in the snail hemolymph is analogous to that of calcium-dependent lectins [37–38]. The snake C-type lectin binds specifically with activated factor X, which functions as anticoagulation factor [39]. FREP1 may also play diverse roles in mosquito biology having similar functions as other FBG-containing proteins in other organisms, and this gene's inactivation could thereby have directly impacted on various developmental processes. In addition, CRISPR/Cas9 mediated gene editing is known for creating unwanted changes at non-target sites though we have selected the gRNA targets with the least potential off-target sites, and such off-target effects might also have contributed to the fitness impairment [40-42], although most off-target mutations would not be expected to become homozygous mutants during the process of the enrichment of a homozygous mosquito population through PCR and sequencing unless they were genetically tightly linked to FREP1. These types of fitness effects are likely to occur after germline-targeted deletion of the majority of Plasmodium host factors, but they

could to some extent be overcome by bloodmeal-inducible adult stage- and tissue-specific gene deletion systems. This challenge could be met through the conditional knockout of *Plasmodium* host factors via blood-meal inducible expression of either the Cas9 protein or gRNAs in the midgut or salivary gland tissues [43]. It is noteworthy that studies with transgenic *Anopheles* that over-express certain anti-*Plasmodium* immune factors in a tissue- and blood-meal-inducible manner do not show measurable fitness effects under laboratory conditions; rather, they show a fitness advantage related to the rapid spread of the transgene in a cage population [25, 44]. However, given the likelihood of mosaic expression of Cas9 protein and gRNA target sequences in different cell types of the same tissue, and the diverse and numerous mutations occurring upon somatic mutation, the effectiveness of conditional CRISPR/Cas9 gene knockout of host factors in blocking the transmission of *Plasmodium* parasites warrants further studies. Nevertheless, the deletion of *Plasmodium* host factors remains a potentially powerful approach for the study of their biological function and ability to influence infection, as well as for the generation of *Plasmodium*-resistant mosquitoes that can compete with wild mosquitoes, and further studies are required to explore this approach.

#### Methods

#### Ethics statement

This study was carried out in strict accordance with the recommendations in the Guide for the Care and Use of Laboratory Animals of the National Institutes of Health. The protocol was approved by the Animal Care and Use Committee of the Johns Hopkins University (permit number MO15H144). Commercial anonymous human blood was used for parasite cultures and mosquito feeding, and informed consent was therefore not applicable. The Johns Hopkins School of Public Health Ethics Committee has approved this protocol.

#### gRNA target design

For the design of guide RNA (gRNA) targets, manually searching the *FREP1* coding sequence for the PAMs (NGG) was initially undertaken, with N representing any nucleotide. We checked candidate gRNA sequences for potential off-target binding using the following web tools: http://zifit.partners.org/ZiFiT/, http://crispr.mit.edu/, http://www.rgenome.net/casdesigner/, and http://chopchop.cbu.uib.no/, and retained three 20-nucleotide specific target sequences that were followed by a PAM and had the fewest predicted genomic off-targets. The final selected gRNA target sequences are within the RNAi primers region and start with a G considering the transcription from U6 promoter starts with a G (S1 Table).

#### Generation of U6-gRNAs constructs via the Golden Gate assembly system

We selected three gRNA targets for the *FREP1* gene, and the forward and reverse primers for each gRNA target DNA sequence (AgFREP1-gRNA1 –F and –R for gRNA1, AgFREP1-gRNA2 –F and –R for gRNA2, and AgFREP1-gRNA3 –F and –R for gRNA3 target) were synthesized by IDT DNA (S1 Table). The two primers of each pair are complementary to each other over the 20 nt specific to the chosen gRNA target and their annealing produces a linker with specific overhangs for the two *Bbs*I sites of the cloning vectors. Three pKSB-sgRNA vectors (pKSB-sgRNA-1, -2, -3) with U6 snRNA polymerase III promoter (AGAP013557), CRISPR RNA invariable sequences, and the RNA polIII TTTTT terminator (U6-Term, Figs 1A and S1) sequences were constructed in a pBluescript cloning vector backbone. U6::gRNA constructs were generated by cloning gRNA target DNA sequences -1, -2, and -3 separately into the *Bbs*I site on these three pKSB-sgRNA vectors followed by sequencing confirmation. At least two

clones from each construct were selected for mini-prep analysis and sequencing to identify the positive clones. The sequencing confirmed three pKSB-FREP1-gRNA plasmids with three gRNA target DNA sequences against *FREP1* were assembled into the destination pDSAT vector through Golden Gate cloning into the *Bsa*I sites, for embryo microinjection of the *A. gambiae* docking line X1 (Fig 1A) [28]. The final plasmid pDSAT-FREP1-gRNA<sub>3</sub>, with 3xP3-TFP (mTurquoise2) eye-specific fluorescence markers, allowed us to screen the positive G<sub>0</sub> (for the transient expression of transgene) and G<sub>1</sub> larvae or mosquitoes by using blue fluorescence (Fig 1B and 1C).

## A. gambiae embryo microinjection and screening for gRNA-expressing transgenic line

The construct (pDSAT-FREP1-gRNA<sub>3</sub>) described above and a helper plasmid carrying the Vasa2::  $\Phi$ C31-Integrase gene (pENTR-R4R3-Vasa2-integrase) [24, 28] were prepared with a Qiagen Endofree Maxi Kit before embryo microinjection according to the established protocols [25, 27–28] (S1 Fig). Freshly laid eggs of docking line X1 were directly aligned against the edge of a nylon membrane that was kept wet with overlaid filter paper soaked with distilled (DI) water, and a mix of the transgenesis plasmid (160 ng/ $\mu$ L) and the integrase helper plasmid (200 ng/µL) in phosphate buffer (0.1 mM NaHPO4 buffer, 5 mM KCl, pH 6.8) was injected into approximately 800 embryos of docking strain X1 [28, 45]. These injected eggs were removed from the slides and kept on a piece of DI water-wetted slanted filter paper (one end of the paper was submerged in the DI water to retain moisture) for 2 days in the insectary before hatching. The hatched G<sub>0</sub> larvae were screened for transient fluorescence expression at the second instar larval (L2) stage by observation with a fluorescence stereomicroscope. Approximately 28% of the hatched larvae showed transient expression of the reporter gene, and the surviving transiently positive  $G_0$  pupae were sexed first (16 females and 12 males) and then organized into two cohorts.  $G_0$  female mosquitoes were crossed with wt (X1) male mosquitoes at a ratio of 1:1, and the surviving  $G_0$  male mosquitoes were crossed with female X1 mosquitoes at a ratio of 1:5. The  $G_1$  progeny were examined for blue fluorescent glowing eyes at both the larval and adult stages (Fig 1B and 1C). FREP1-gRNA-expressing transgenic positive progeny was outcrossed with X1 for two generations followed by two generations of self-crossing to enrich the homozygous mosquitoes by screening the fluorescence in the larvae before crossing with the Vasa-Cas9 strain to generate germ-line knockout mutants (S1 Fig).

## Generation of CRISPR/Cas9 knockout mutants and molecular confirmation of CRISPR activity at the target loci

The G<sub>4</sub> (fourth-generation) FREP1-gRNA lines were crossed with mosquitoes expressing the Cas9 protein in the germ-line (*Vasa2-Cas9*; with yellow (YFPvenus) fluorescence in the eyes) [16] (S1 Fig). We crossed about 100 virgin FREP1-gRNA female mosquitoes with 50 male *Vasa-Cas9* mosquitoes to generate knockout mutants. Since the parental gRNA mosquitoes were not fully homozygous, we eliminated all the larvae that did not carry the FREP1-gRNA sequences by retaining only larvae with both transgenes (*FREP1-gRNA/Cas9* with both blue and YFP fluorescence [Fig 1D]). The males from this transheterozygous *FREP1-gRNA/Cas9* (FREP1/Cas9) culture were then crossed with wt virgin females. Among the progeny of this new cross, only the virgin females (potential heterozygous mutants) were selected and crossed with wt males. This final cross was then fed on naïve mice, and 3 days post-blood feeding, each individual female mosquito was placed in a single tube (50 ml Falcon tube with 1 cm of water and a conical Whatman filter surrounding the plastic walls) for these female

mosquitoes to lay eggs. Two days later, each female that produced eggs was collected in a single Eppendorf tube with 100 µl of lysis buffer for genomic DNA (gDNA) extraction. The mutant region from the mother was amplified by PCR with flanking primers (S1 Table). PCR products were run on a 1.2% agarose gel to visualize both the mutant band (with a deletion between FREP1-gRNA1 and FREP1-gRNA2 targets) and the wt band, since the mothers were heterozygous. PCR products were also sequenced to confirm the deletion (SI S1 File). The progenies from each mother were kept as separate families. Among the 50 founder mothers, 5 showed the same 155 bp deletion mutation, and 12 (sub-grouped into 4 different sequence variations) showed smaller deletions which were confirmed by sequencing. After sequencing confirmation of the mutants, the separate progenies from mutant mothers were selected and sexed at the pupal stage to separate males and females before mating. PCR on one leg (Phire kit with dilution protocol) was used to identify the mutant mosquitoes, and all the mutant mosquitoes with the same genotype were pooled. This process was repeated in the next generation to enrich for homozygous mutants. gDNA was prepared as previously described [25, 27]. PCR with FREP1-KO-seqF1 and FREP1-KO-seqR3 primers (S1 Table, S1 Fig) was used to amplify a 478 bp fragment for sequencing. The mutation of the gene was determined by alignment of the mutant sequence with the wt sequence through Blast2 (S1 File).

#### P. falciparum and P. berghei infection assays

To determine the anti-Plasmodium activity, the FREP1 knockout mutants and wt X1, Vasa-Cas9 (Cas9), and FREP1-gRNA lines were fed on NF54 P. falciparum gametocyte cultures (provided by the Johns Hopkins Malaria Institute Core Facility; NF54 obtained from MR4) through artificial membranes at 37°C or on a P. berghei wt ANKA 2.34-infected Swiss Webster mouse (at 19°C) [18, 25]. The adult mosquitoes were starved for 3 to 5 h prior to feeding to ensure engorgement. Unfed mosquitoes were removed immediately after blood feeding, and the remaining were incubated for a further 8 days at 27°C or 13 days at 19°C for P. falciparum and P. berghei, respectively. Midguts were dissected in PBS, stained with 0.2% mercurochrome, and examined using a phase-contrast microscope (Leica). At least two to three biological replicates were performed for each experiment. As described previously, after comparing the variance in oocyst numbers between different replicates, equal numbers of midguts from the various replicates were selected randomly and pooled for producing dot-plots using GraphPad Prism 7 software [25, 46-47]. A published method [25, 48] was used, with some modifications, to determine the sporozoite loads in the salivary glands of the infected mosquitoes. Salivary glands were dissected, and individual glands were placed in Eppendorf tubes with 30  $\mu$ l of PBS, followed by homogenization on ice. A sample (10 µl) of this homogenate was placed in a Neubauer counting chamber and counted after 5 min using a Leica phase-contrast microscope at 400x magnification.

The dot plots of oocyst and sporozoite numbers in each individual gut and salivary gland were generated using GraphPad Prism 7 software, along with the median values, represented as red bars. The p-values were calculated using a non-parametric Mann-Whitney test or Kruskal-Wallis ANOVA on ranks and used to determine the significance of differential infection levels as described in [18, 25, 46]. Chi-squared analysis was used to determine the significance of infection prevalence [29, 46]. To gain a more detailed picture of prevalence and infection intensity at both high and low infection intensity levels, we have separated these two attributes into two data sets for the statistical analysis, and calculated p-values with or without uninfected mosquitoes as shown in S2 and S3 Tables. Median, range of infection, and N numbers are presented in S2 and S3 Tables.

## Wing length, pupation time, life span, fecundity, and egg hatching rate measurements

Adult wing length (both male and female) was used as a surrogate measurement for mosquito size. Mosquitoes were anesthetized on ice and kept on a cold plate for wing length measurement. Wing length was measured manually from the distal end of the alula to the tip of the wing (without the hairy fringe) through a microscope objective containing a scale bar calibrated to a 1-mm stage micrometer without taking pictures or using software [44].

Larvae were maintained according to a standard procedure with 400 larvae in each tray, and the number of pupae present each day was recorded on a daily basis to determine development time [44]. At least three biological replicates were included for each strain.

In order to measure the life span of the various mosquito lines, adult mosquitoes were placed into cups within 12 h of emergence with a cotton pad constantly impregnated with a 10% sucrose solution. They were held there until all the mosquitoes in that cup died, and the number of dead mosquitoes in the cup was recorded and the dead mosquitoes removed daily [25, 44]. For the life span assays of mosquitoes receiving a single blood meal (blood-fed on mice), the mosquitoes were offered a blood meal 5 days after emergence, and only mosquitoes taking a blood meal on day 5 were kept for the rest of the study. The survival percentage represents the mean survival percentage for all three biological replicates of 35 mosquitoes each. Statistical significance was determined by Kaplan-Meier survival analysis with pooled data from three replicates by using GraphPad Prism 7 software, and *p*-values were determined by Wilcoxon test as described in [25, 29].

For the fecundity assay, approximately 50 7-day-old adult female wt X1, *Vasa-Cas9*, FREP1-gRNA, and heterozygous or homozygous *FREP1* knockout mutant mosquitoes were allowed to feed on mice. Mosquitoes were anesthetized on ice immediately following the blood meal, and all non-engorged mosquitoes were discarded. At 2 days post-blood feeding, female mosquitoes were separated into individual vials (50 ml Falcon tubes) containing moist filter paper with 1 ml of water on the bottom and allowed to oviposit, and the number of eggs laid by each female was recorded 4 days after the blood feeding using light microscopy. The females that died before laying eggs were excluded from the assays. After each count, the eggs were submerged in larval rearing water in the same individual tubes to allow eggs to hatch. First-and second- instar larvae were counted under a light microscope. The fecundity and larval hatch-rate assays were performed for two consecutive generations (two biological replicates), and the number of eggs laid by each female and their hatch rates were pooled to calculate the median value. Statistical significance was determined using the Mann-Whitney test with GraphPad Prism 7 software. The control mosquitoes were reared under the same conditions as the mutants.

#### Supporting information

**S1 Fig. Generation of** *FREP1-gRNA*-expressing transgenic line and homozygous *FREP1* **knockout mutants.** The generation of *FREP1-gRNA*-expressing transgenic line and homozygous *FREP1* knockout mutants is outlined. (TIF)

**S1 Table. List of gRNA target sequences and primers used in the study.** Primers used for generation of constructs for embryo microinjections and the verification of transgene integration and gene knockout. (DOCX)

**S2 Table.** *P. falciparum* infection data. Additional statistical analysis of numbers of *P. falciparum* parasites for Fig 2 (Mann-Whitney, Kruskal-Wallis test, chi-squared test). (DOCX)

**S3 Table.** *P. berghei* infection data. Additional statistical analysis of numbers of *P. berghei* parasites for Fig 3. (Mann-Whitney, Kruskal-Wallis test, chi-squared test). (DOCX)

**S1 File. Sequences.** Sequences related to Fig <u>1F</u> and *FREP1* Knockouts. (DOCX)

#### Acknowledgments

We would like to thank the Johns Hopkins Malaria Research Institute Insectary and Parasitology core facilities, and especially Dr. Mlambo for providing *P. berghei*-infected mice and Dr. Tripathi for providing *P. falciparum* gametocyte cultures. We thank Dr. Jun Li for sharing FREP1 antibody. We also thank Dr. Deborah McClellan for editorial assistance.

#### **Author Contributions**

Conceptualization: Eric Marois, George Dimopoulos.

Formal analysis: Yuemei Dong.

Investigation: Yuemei Dong, Maria L. Simões, Eric Marois.

Methodology: Yuemei Dong, Maria L. Simões, Eric Marois.

Project administration: George Dimopoulos.

Supervision: Eric Marois, George Dimopoulos.

Validation: Yuemei Dong.

Writing - original draft: Yuemei Dong, George Dimopoulos.

Writing - review & editing: Eric Marois, George Dimopoulos.

#### References

- Murray CJ, Rosenfeld LC, Lim SS, Andrews KG, Foreman KJ, Haring D, et al. Global malaria mortality between 1980 and 2010: a systematic analysis. Lancet. 2012; 379(9814):413–31. https://doi.org/10. 1016/S0140-6736(12)60034-8 PMID: 22305225.
- 2. Ghosh A, Edwards MJ, Jacobs-Lorena M. The journey of the malaria parasite in the mosquito: hopes for the new century. Parasitol Today. 2000; 16(5):196–201. PMID: <u>10782078</u>.
- Wang S, Jacobs-Lorena M. Genetic approaches to interfere with malaria transmission by vector mosquitoes. Trends Biotechnol. 2013; 31(3):185–93. https://doi.org/10.1016/j.tibtech.2013.01.001 PMID: 23395485.
- Sreenivasamurthy SK, Dey G, Ramu M, Kumar M, Gupta MK, Mohanty AK, et al. A compendium of molecules involved in vector-pathogen interactions pertaining to malaria. Malar J. 2013; 12:216. <u>https:// doi.org/10.1186/1475-2875-12-216 PMID: 23802619</u>.
- Osta MA, Christophides GK, Kafatos FC. Effects of mosquito genes on Plasmodium development. Science. 2004; 303(5666):2030–2. https://doi.org/10.1126/science.1091789 PMID: 15044804.
- Christophides GK. Transgenic mosquitoes and malaria transmission. Cell Microbiol. 2005; 7(3):325– 33. https://doi.org/10.1111/j.1462-5822.2005.00495.x PMID: 15679836.
- Mendes AM, Schlegelmilch T, Cohuet A, Awono-Ambene P, De Iorio M, Fontenille D, et al. Conserved mosquito/parasite interactions affect development of Plasmodium falciparum in Africa. PLoS Pathog. 2008; 4(5):e1000069. https://doi.org/10.1371/journal.ppat.1000069 PMID: 18483558.

- Dinglasan RR, Kalume DE, Kanzok SM, Ghosh AK, Muratova O, Pandey A, et al. Disruption of Plasmodium falciparum development by antibodies against a conserved mosquito midgut antigen. Proc Natl Acad Sci U S A. 2007; 104(33):13461–6. https://doi.org/10.1073/pnas.0702239104 PMID: 17673553.
- Basu S, Aryan A, Overcash JM, Samuel GH, Anderson MA, Dahlem TJ, et al. Silencing of end-joining repair for efficient site-specific gene insertion after TALEN/CRISPR mutagenesis in Aedes aegypti. Proc Natl Acad Sci U S A. 2015; 112(13):4038–43. <u>https://doi.org/10.1073/pnas.1502370112</u> PMID: 25775608.
- Dong S, Lin J, Held NL, Clem RJ, Passarelli AL, Franz AW. Heritable CRISPR/Cas9-mediated genome editing in the yellow fever mosquito, Aedes aegypti. PLoS One. 2015; 10(3): e0122353. https://doi.org/ 10.1371/journal.pone.0122353 PMID: 25815482.
- Kistler KE, Vosshall LB, Matthews BJ. Genome engineering with CRISPR-Cas9 in the mosquito Aedes aegypti. Cell Rep. 2015; 11(1):51–60. https://doi.org/10.1016/j.celrep.2015.03.009 PMID: 25818303.
- Hall AB, Basu S, Jiang X, Qi Y, Timoshevskiy VA, Biedler JK, et al. SEX DETERMINATION. A maledetermining factor in the mosquito Aedes aegypti. Science. 2015; 348(6240):1268–70. https://doi.org/ 10.1126/science.aaa2850 PMID: 25999371.
- Itokawa K, Komagata O, Kasai S, Ogawa K, Tomita T. Testing the causality between CYP9M10 and pyrethroid resistance using the TALEN and CRISPR/Cas9 technologies. Sci Rep. 2016; 6:24652. https://doi.org/10.1038/srep24652 PMID: 27095599.
- Grigoraki L, Puggioli A, Mavridis K, Douris V, Montanari M, Bellini R, et al. Striking diflubenzuron resistance in Culex pipiens, the prime vector of West Nile Virus. Sci Rep. 2017; 7(1):11699. https://doi.org/ 10.1038/s41598-017-12103-1 PMID: 28916816.
- Gantz VM, Jasinskiene N, Tatarenkova O, Fazekas A, Macias VM, Bier E, et al. Highly efficient Cas9mediated gene drive for population modification of the malaria vector mosquito Anopheles stephensi. Proc Natl Acad Sci U S A. 2015; 112(49):E6736–43. <u>https://doi.org/10.1073/pnas.1521077112</u> PMID: 26598698.
- Hammond A, Galizi R, Kyrou K, Simoni A, Siniscalchi C, Katsanos D, et al. A CRISPR-Cas9 gene drive system targeting female reproduction in the malaria mosquito vector Anopheles gambiae. Nat Biotechnol. 2016; 34(1):78–83. https://doi.org/10.1038/nbt.3439 PMID: 26641531.
- Hammond AM, Kyrou K, Bruttini M, North A, Galizi R, Karlsson X, et al. The creation and selection of mutations resistant to a gene drive over multiple generations in the malaria mosquito. PLoS Genet. 2017; 13(10):e1007039. https://doi.org/10.1371/journal.pgen.1007039 PMID: 28976972.
- Dong Y, Aguilar R, Xi Z, Warr E, Mongin E, Dimopoulos G. Anopheles gambiae immune responses to human and rodent Plasmodium parasite species. PLoS Pathog. 2006; 2(6):e52. <u>https://doi.org/10. 1371/journal.ppat.0020052</u> PMID: 16789837.
- Dong Y, Dimopoulos G. Anopheles fibrinogen-related proteins provide expanded pattern recognition capacity against bacteria and malaria parasites. J Biol Chem. 2009; 284(15):9835–44. https://doi.org/ 10.1074/jbc.M807084200 PMID: 19193639.
- Simoes ML, Dong Y, Hammond A, Hall A, Crisanti A, Nolan T, et al. The Anopheles FBN9 immune factor mediates Plasmodium species-specific defense through transgenic fat body expression. Dev Comp Immunol. 2017; 67:257–65. https://doi.org/10.1016/j.dci.2016.09.012 PMID: 27667688.
- Zhang G, Niu G, Franca CM, Dong Y, Wang X, Butler NS, et al. Anopheles Midgut FREP1 Mediates Plasmodium Invasion. J Biol Chem. 2015; 290(27):16490–501. <u>https://doi.org/10.1074/jbc.M114</u>. 623165 PMID: 25991725.
- Niu G, Wang B, Zhang G, King JB, Cichewicz RH, Li J. Targeting mosquito FREP1 with a fungal metabolite blocks malaria transmission. Sci Rep. 2015; 5:14694. https://doi.org/10.1038/srep14694 PMID: 26437882.
- Niu G, Franc AC, Zhang G, Roobsoong W, Nguitragool W, Wang X, et al. The fibrinogen-like domain of FREP1 protein is a broad-spectrum malaria transmission-blocking vaccine antigen. J Biol Chem. 2017; 292(28):11960–9. https://doi.org/10.1074/jbc.M116.773564 PMID: 28533429.
- Papathanos PA, Windbichler N, Menichelli M, Burt A, Crisanti A. The vasa regulatory region mediates germline expression and maternal transmission of proteins in the malaria mosquito Anopheles gambiae: a versatile tool for genetic control strategies. BMC Mol Biol. 2009; 10:65. https://doi.org/10.1186/ 1471-2199-10-65 PMID: 19573226.
- Dong Y, Das S, Cirimotich C, Souza-Neto JA, McLean KJ, Dimopoulos G. Engineered anopheles immunity to Plasmodium infection. PLoS Pathog. 2011; 7(12):e1002458. <u>https://doi.org/10.1371/journal.ppat.1002458</u> PMID: 22216006.
- Lobo NF, Clayton JR, Fraser MJ, Kafatos FC, Collins FH. High efficiency germ-line transformation of mosquitoes. Nat Protoc. 2006; 1(3):1312–7. https://doi.org/10.1038/nprot.2006.221 PMID: 17406416.

- Dong Y, Cirimotich CM, Pike A, Chandra R, Dimopoulos G. Anopheles NF-kappaB-regulated splicing factors direct pathogen-specific repertoires of the hypervariable pattern recognition receptor AgDscam. Cell Host Microbe. 2012; 12(4):521–30. https://doi.org/10.1016/j.chom.2012.09.004 PMID: 23084919.
- Volohonsky G, Terenzi O, Soichot J, Naujoks DA, Nolan T, Windbichler N, et al. Tools for Anopheles gambiae Transgenesis. G3 (Bethesda). 2015; 5(6):1151–63. <u>https://doi.org/10.1534/g3.115.016808</u> PMID: 25869647.
- Corby-Harris V, Drexler A, Watkins de Jong L, Antonova Y, Pakpour N, Ziegler R, et al. Activation of Akt signaling reduces the prevalence and intensity of malaria parasite infection and lifespan in Anopheles stephensi mosquitoes. PLoS Pathog. 2010; 6(7):e1001003. <u>https://doi.org/10.1371/journal.ppat.</u> 1001003 PMID: 20664791.
- Eckhoff PA, Wenger EA, Godfray HC, Burt A. Impact of mosquito gene drive on malaria elimination in a computational model with explicit spatial and temporal dynamics. Proc Natl Acad Sci U S A. 2017; 114 (2):E255–E64. https://doi.org/10.1073/pnas.1611064114 PMID: 28028208.
- Sinkins SP, Gould F. Gene drive systems for insect disease vectors. Nat Rev Genet. 2006; 7(6):427– 35. https://doi.org/10.1038/nrg1870 PMID: 16682981.
- Blagborough AM, Churcher TS, Upton LM, Ghani AC, Gething PW, Sinden RE. Transmission-blocking interventions eliminate malaria from laboratory populations. Nat Commun. 2013; 4:1812. <u>https://doi.org/10.1038/ncomms2840 PMID: 23652000.</u>
- Godfray HCJ, North A, Burt A. How driving endonuclease genes can be used to combat pests and disease vectors. BMC Biol. 2017; 15(1):81. https://doi.org/10.1186/s12915-017-0420-4 PMID: 28893259.
- Rodgers FH, Gendrin M, Wyer CAS, Christophides GK. Microbiota-induced peritrophic matrix regulates midgut homeostasis and prevents systemic infection of malaria vector mosquitoes. PLoS Pathog. 2017; 13(5):e1006391. https://doi.org/10.1371/journal.ppat.1006391 PMID: 28545061.
- Lu J, Teh C, Kishore U, Reid KB. Collectins and ficolins: sugar pattern recognition molecules of the mammalian innate immune system. Biochim Biophys Acta. 2002; 1572(2–3):387–400. PMID: 12223281.
- Fujita T. Evolution of the lectin-complement pathway and its role in innate immunity. Nat Rev Immunol. 2002; 2(5):346–53. https://doi.org/10.1038/nri800 PMID: 12033740.
- Zhang SM, Nian H, Zeng Y, Dejong RJ. Fibrinogen-bearing protein genes in the snail Biomphalaria glabrata: Characterization of two novel genes and expression studies during ontogenesis and trematode infection. Dev Comp Immunol. 2008; 32(10):1119–30. https://doi.org/10.1016/j.dci.2008.03.001 PMID: 18417215.
- Zhang SM, Zeng Y, Loker ES. Expression profiling and binding properties of fibrinogen-related proteins (FREPs), plasma proteins from the schistosome snail host Biomphalaria glabrata. Innate Immun. 2008; 14(3):175–89. https://doi.org/10.1177/1753425908093800 PMID: 18562576.
- Zhang Y, Xu X, Shen D, Song J, Guo M, Yan X. Anticoagulation factor I, a snaclec (snake C-type lectin) from Agkistrodon acutus venom binds to FIX as well as FX: Ca2+ induced binding data. Toxicon. 2012; 59(7–8):718–23. https://doi.org/10.1016/j.toxicon.2012.03.006 PMID: 22445822.
- Chen JS, Dagdas YS, Kleinstiver BP, Welch MM, Sousa AA, Harrington LB, et al. Enhanced proofreading governs CRISPR-Cas9 targeting accuracy. Nature. 2017. <u>https://doi.org/10.1038/nature24268</u> PMID: 28931002.
- Frock RL, Hu J, Meyers RM, Ho YJ, Kii E, Alt FW. Genome-wide detection of DNA double-stranded breaks induced by engineered nucleases. Nat Biotechnol. 2015; 33(2):179–86. <u>https://doi.org/10.1038/nbt.3101</u> PMID: 25503383.
- Fu Y, Foden JA, Khayter C, Maeder ML, Reyon D, Joung JK, et al. High-frequency off-target mutagenesis induced by CRISPR-Cas nucleases in human cells. Nat Biotechnol. 2013; 31(9):822–6. https://doi. org/10.1038/nbt.2623 PMID: 23792628.
- Xue Z, Wu M, Wen K, Ren M, Long L, Zhang X, et al. CRISPR/Cas9 mediates efficient conditional mutagenesis in Drosophila. G3 (Bethesda). 2014; 4(11):2167–73. <u>https://doi.org/10.1534/g3.114</u>. 014159 PMID: 25193494.
- Pike A, Dong Y, Dizaji NB, Gacita A, Mongodin EF, Dimopoulos G. Changes in the microbiota cause genetically modified Anopheles to spread in a population. Science. 2017; 357(6358):1396–9. <u>https://</u> doi.org/10.1126/science.aak9691 PMID: 28963254.
- Meredith JM, Underhill A, McArthur CC, Eggleston P. Next-generation site-directed transgenesis in the malaria vector mosquito Anopheles gambiae: self-docking strains expressing germline-specific phiC31 integrase. PLoS One. 2013; 8(3):e59264. https://doi.org/10.1371/journal.pone.0059264 PMID: 23516619.
- **46.** Mitri C, Jacques JC, Thiery I, Riehle MM, Xu J, Bischoff E, et al. Fine pathogen discrimination within the APL1 gene family protects Anopheles gambiae against human and rodent malaria species. PLoS Pathog. 2009; 5(9):e1000576. https://doi.org/10.1371/journal.ppat.1000576 PMID: 19750215.

- Kumar S, Molina-Cruz A, Gupta L, Rodrigues J, Barillas-Mury C. A peroxidase/dual oxidase system modulates midgut epithelial immunity in Anopheles gambiae. Science. 2010; 327(5973):1644–8. https://doi.org/10.1126/science.1184008 PMID: 20223948.
- Cirimotich CM, Dong Y, Clayton AM, Sandiford SL, Souza-Neto JA, Mulenga M, et al. Natural microbemediated refractoriness to Plasmodium infection in Anopheles gambiae. Science. 2011; 332 (6031):855–8. https://doi.org/10.1126/science.1201618 PMID: 21566196.